Author's response to reviews

Title: Cholera outbreaks (2012) in three districts of Nepal reveal clonal transmission of multi-drug resistant Vibrio cholerae O1

Authors:

Sameer M. Dixit (s.dixit@cmdn.org)
Fatema-Tuz Johura (mbmita@gmail.com)
Sulochana Manandhar (manandhar.sulochana@outlook.com)
Abdus Sadique (sadique@icddrb.org)
Rajesh M. Rajbhandari (r.rajbhandari@cmdn.org)
Shahnewaj B. Mannan (newaj_mb@yahoo.com)
Mahamud-ur Rashid (mahamudur@icddrb.org)
Saiful Islam (saiful_mbd@yahoo.com)
Dibesh Karmacharya (dibesh@cmdn.org)
Haruo Watanabe (haruwata@nih.go.jp)
R. B. Sack (rsack@jhsph.edu)
Alejandro Cravioto (Alejandro.Cravioto@ivi.int)
Munirul Alam (munirul@icddrb.org)

Version: 6  Date: 8 July 2014

Author's response to reviews: see over
To
Editor,
BMC Infectious disease

Subject: Fourth version of the revised manuscript for publication

Dear Sir/M’m,
Please find the fourth revision of the manuscript (MS) entitled “Cholera outbreaks (2012) in three districts of Nepal reveal clonal transmission of multi-drug resistant Vibrio cholerae O1” for your kind consideration. We have added the information that the editor suggested.

We have deposited the phylogenetic data in Dryad Digital Repository and mentioned the DOI link in the manuscript

May I therefore request you to kindly expedite the publication which is long due for us?

Sincerely yours,

Munirul Alam